Author's response to reviews

Title: Servelle-Martorell Syndrome With Extensive Upper Limb Involvement - A Case Report

Authors:

Raju Karuppal Dr (drrajuortho@rediffmail.com)
Rajendran V R Professor (rajendran@hotmail.com)
Brijesh P Valsalan Dr (brijeshpv@gmail.com)
Gopakumar T S Professor (gopats@sifymail.com)
Chathoth Meethal Kumaran Professor (cmk5303@yahoo.co.in)
Chembu Kara Vasu Professor (ckvasu2000@rediffmail.com)

Version: 2 Date: 29 October 2007

Author's response to reviews: see over
Dear Sir,

We are submitting a rare presentation of Servelle-Martorell syndrome, with extensive involvement of right upper limb and perscapular region. Cases of only minimal upper limb involvement have been described in the English literature. This article stresses the need for suspecting Servelle-Martorell syndrome as one of the causes of angiodysplastic limb hypertrophy and the rarity of its extensive nature of presentation.

We request you to consider this article for presentation in your esteemed journal.

Thanking you.

Yours truly,

Raju Karuppal